Author's response to reviews

Title: The use of Ligasure Vessel Sealing System in Ivor Lewis Esophagectomy

Authors:

Fuat Sayir (sayirfuat@yahoo.com)
Ufuk çobanoğlu (drucobanoglu@hotmail.com)
Abidin Şehitoğulları (abidin_sehitoglu@yahoo.com)

Version: 3 Date: 20 January 2012

Author's response to reviews:

The details of my revised manuscript as follows:

Major revisions: 1. The title of article is as follow: The use of LigaSure Vessel Sealing System in Ivor Lewis Esophagectomy.
2. The abstract in the manuscript and the abstract in the online submission system are identical.
3. The competing interests section written as: The authors declare that they have no competing interests.
4. The authors' contributions were written in the section of Author's Contributions.

Minor Revisions were made.